Author’s response to reviews

Title: (Correcting) Misdiagnoses of Asthma: A Cost Effectiveness Analysis

Authors:

Smita Pakhale (spakhale@ohri.ca)
Amanda Sumner (ASumner@cheo.on.ca)
Douglas Coyle (dcoyle@uottawa.ca)
Katherine Vandemheem (kvandemheem@toh.on.ca)
Shawn Aaron (saaron@ohri.ca)

Version: 2 Date: 9 November 2010

Author’s response to reviews:

Dear Sir/madam:

We are submitting an original article entitled, “(Correcting) misdiagnoses of asthma: a cost effectiveness analysis” to your esteemed journal.

Though asthma epidemic is growing, misdiagnosis of asthma is quite common. This study sheds light on cost advantages of making accurate diagnosis of asthma, even after many years since the first diagnosis was made.

I hope you find this article as exciting and interesting as we do.

Thank you.

Sincerely,

Dr. Smita Pakhale